Title: Emotionally induced galactorrhoea in a non-lactating female - "Pseudo-Lactation"?

Authors:

Prasad Katulanda (prasad.katulanda@yahoo.com)
Chinthana Dematapitiya (chinthanadematapitiya@yahoo.com)
Harsha Dissanayake (dissanayakeha@gmail.com)
Sisil Keerthisena (sisilkeerthisena@yahoo.com)

Version: 3
Date: 25 November 2014

Author's response to reviews: see over
November 24, 2014

Timothy Shipley  
Executive Editor  
The BMC Endocrine Disorders  
The BioMed Central

Dear Timothy Shipley,

Revised Manuscript submission MS: 8252377531356717

Title: Emotionally induced galactorrhoea in a non-lactating female – “Pseudo- Lactation”?  
Thank you for your decision to accept our manuscript on emotionally induced galactorrhoea in a non-lactating female – “Pseudo- Lactation”? to be published in your esteemed journal.

We are pleased to submit our revised manuscript for your consideration to be published in the BMC endocrine disorders.

We have addressed all the concerns by referee 1, referee 2 and editor’s additional request of BioMed Central in our revised manuscript and have responded in this cover later as a point by point response.

Referee 1

Comment

I am satisfied with the changes regarding the following manuscript.

Response

We are thankful for the comments. So no further changes needed according to the comment regarding revised manuscript.
Referee 2

Comment

Unfortunately, the provided data do not unequivocally support the case. The investigations are inadequate and there are many assumptions and less evidence in order to establish causal association of this patient’s galactorrhea, with the visual or auditory stimuli from the newborn.

Response

We agree with the reviewers comments and addressed these issues as strengths and weaknesses. However since this case illustrates a novel hypothesis and phenomenon we are thankful to the ditors for consideration to publish the manuscript. We believe this publication will help more people to look into this possibility in future cases.

Editor's Additional Request

(1) Requesting for Email-add

E – Mail addresses of all co-authors have been added to title page of new revised manuscript.

(2) Please remove the Disclosure Statement and Acknowledgement in the Title Page.

Disclosure statement and acknowledgement has been removed from new revised manuscript.

We hope that we have satisfactory addressed major and minor concerns of both referees and editor’s additional request in our revised manuscript.

We feel the reason for paucity of literature on this may be due to the fact it would have been overlooked as a cause of galactorrhoea.
Therefore through this manuscript we may be able to unveil a cause of galactorrhoea and stimulate more research on the pathophysiology of his mechanism.

The authors declare no competing interests related to this paper.

Thank you for considering the paper for publication.

Yours sincerely

Dr. Prasad Katulanda, MD, FRCP (Lond), FCCP, DPhil (Oxon)
Senior Lecturer in Medicine
University of Colombo
Hon. Consultant Endocrinologist & Diabetologist
University Medical Unit National Hospital, Sri Lanka
Tel: 0112695300 Ext 318
Mobile: +94772920991
President, Endocrine Society of Sri Lanka